Author's response to reviews

Title: The Dutch version of the Oral Health Impact Profile (OHIP-NL): Translation, reliability and construct validity

Authors:

Marylee J van der Meulen (mvdmeulen@acta.nl)
Mike T John (Mike.John@medizin.uni-leipzig.de)
Machiel Naeije (M.Naeije@acta.nl)
Frank Lobbezoo (F.Lobbezoo@acta.nl)

Version: 4 Date: 7 April 2008

Author's response to reviews: see over
Dear sir/madam,

Thank you for accepting our article about the Dutch version of the OHIP. We have made one change, namely including the 4 tables in the text, as you suggested, and deleting them from the separate boxes. We did not include any acknowledgments or funds, since there weren’t any.

I hope this was done correctly. If any changes are still necessary, please let me know.

Sincerely,

Marylee van der Meulen